Author's response to reviews

Title: Failure of recombinant factor VIIa in a patient with severe polymicrobial sepsis and postoperative uncontrolled intraabdominal bleeding

Authors:

Anna Conen (conena@uhbs.ch)
Maja Weisser (weisserm@uhbs.ch)
Dimitrios A Tsakiris (tsakirisd@uhbs.ch)
Martin Siegemund (siegemundm@uhbs.ch)

Version: 6 Date: 12 March 2007

Author's response to reviews: see over
Dear Ladies and Gentlemen

Please find enclosed the definite version of our manuscript “Failure of recombinant factor VIIa in a patient with severe polymicrobial sepsis and postoperative uncontrolled intraabdominal bleeding” with number MS 1054361595119854 to “BMC Infectious Diseases”.
We did all the formatting changes you were asking for.

Yours faithfully

Anna Conen, MD